Author’s response to reviews

Title: An extensible framework and database of infectious disease for biosurveillance

Authors:

Ashlynn Daughton (adaughton@lanl.gov)
Reid Priedhorsky (reidpr@lanl.gov)
Geoffrey Fairchild (gfairchild@lanl.gov)
Nicholas Generous (generous@lanl.gov)
Andrea Hengartner (ahengartner@lanl.gov)
Esteban Abeyta (esteban_abeyta@lanl.gov)
Nileena Velappan (nileena@lanl.gov)
Antonietta Lillo (alillo@lanl.gov)
Karen Stark (karens@digitalinfuzion.com)
Alina Deshpande (deshpande_a@lanl.gov)

Version: 3 Date: 26 Jul 2017

Author’s response to reviews:

Requisite files to correctly compile LaTex should be uploaded now. Many thanks for you help figuring out the problem.